Author's response to reviews

Title: Developing and testing an instrument to measure the presence of conditions for successful implementation of quality improvement collaboratives

Authors:

Michel LA Duckers (m.duckers@nivel.nl)
Cordula Wagner (c.wagner@nivel.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)

Version: 4 Date: 2 August 2008

Author's response to reviews:

Dear Editor,

In the final manuscript the following changes were made:
- removing reference to the funding agency from title page;
- email adresses of other authors were added to title page;
- changed methodology into methods;
- removed table placement texts throughout the manuscript;
- added missing author names in reference list, removed et al.;
- made sure table sizes fit into a A$ prtrait page.

Kind regards,

Michel Dückers